FEATURED STUDENT RESEARCH PAPER



Restricted use of space in an endemic lizard of the Andes: addressing the effects of intrinsic and environmental factors

Ana E. Victorica Erostarbe^{1,2} · Gustavo A. Fava^{1,2} · Juan C. Acosta¹

Received: 11 August 2021 / Revised: 11 December 2021 / Accepted: 14 December 2021 / Published online: 8 January 2022 © The Author(s), under exclusive licence to Springer-Verlag GmbH Germany, part of Springer Nature 2022

Abstract

Spatial ecology often results from the interaction between animals' spatial use behavior and the characteristics of their environment, which can have implications for social relationships between individuals. Therefore, understanding individual spatial use is crucial to species conservation as it directly impacts population stability and species persistence. *Phymaturus* is a vulnerable genus of liolaemid lizards. In this study, we address the patterns involved in the variation of spatial use of saxicolous lizards, focusing on the endemic *Phymaturus williamsi*. We used generalized linear models to evaluate factors (intrinsic and extrinsic) that may affect lizards' use of space (home range and microhabitat). Home ranges estimated for population varied from 0.0063 to 44.36 m². This species presented the smallest home range recorded to date among Liolaemidae. Home range size varied seasonally and was not related to lizard snout-vent length. Female home ranges were significantly smaller than male and juvenile home ranges. An interesting fact is that home range size was not related to microenvironmental variables: wind speed, distance to vegetation, microhabitat used, thermoregulation condition, or shelter condition. However, home range size and microhabitat use were related to substrate temperature. This species specialized in two rock types (basaltic and feldspathic rock). We conclude that *P. williamsi* is a lizard with strict and reduced microhabitats, dependent on substrate temperature, which makes it highly vulnerable to habitat changes and requires careful habitat management. To our knowledge, this is the first study of its kind on this species and has important implications for its conservation.

Significance statement

Spatial ecology affects and is affected by almost every aspect of an individual's biology and is therefore crucial for conservation. By studying the spatial ecology of individuals during a complete activity period (from spring to autumn), we were able to obtain valuable information on how the use of space changes as the associated factors, both intrinsic or extrinsic, change for an individual. Studying home range and microhabitat use, we observe that the spatial ecology of *Phymaturus williamsi* responds to intrapopulation and seasonal and substrate temperature variables, with these being very specific for utilization of microhabitat. Therefore, we propose that this species could be threatened, due to the combination of aspects analyzed in this study and its biology. We highlight that spatial ecology provides comprehensive information that must be taken into account for the conservation and management of species and their habitats.

Keywords Argentina · Home range · Microhabitat use · Spatial use · Overlap · Phymaturus williamsi

Communicated by S. Joy Downes

Ana E. Victorica Erostarbe anivictorica@gmail.com

¹ Present Address: Gabinete de Diversidad y Biología de Vertebrados del Árido (DIBIOVA), Departamento de Biología, Facultad de Ciencias Exactas, Físicas y Naturales, Universidad Nacional de San Juan, Av. Ignacio de la Roza 590 (O), Rivadavia, CPA: J5402DCS San Juan, Argentina

² CONICET (Consejo Nacional de Investigaciones Científicas y Técnicas), Buenos Aires, Argentina

Introduction

Spatial ecology often results from the interaction between animals' spatial use behavior and the characteristics of their environment (climate, food, predators, diseases, and other abiotic and biotic environmental factors), which can have implications for social relationships between individuals (Pough et al. 2001; Hoffmann et al. 2015; He et al. 2019; Shaw 2020). As a result, different distributions of these environmental factors in space and time will translate into different spatial and temporal availability of potential mates and the ability to monopolize them (Hoffmann et al. 2015). Each of these intraspecific interactions and abiotic factors is filtered through the individual to shape its spatial ecology (Shaw 2020). Understanding individual spatial use is crucial for species conservation, as it directly impacts population stability and species persistence (Evans et al. 2019). For example, in order to predict the consequences of anthropogenic change, it is essential to understand the capacity and motivation for the spatial use of species within complex landscapes (Evans et al. 2019). Consequently, spatial ecology affects and is affected by almost every aspect of an individual's biology (Pough et al. 2001; Shaw 2020) and reflects many fundamental ecological processes (Ousterhout and Burkhart 2017).

A relevant factor in spatial ecology is home range (hereafter, HR), defined as the total area within which an individual moves and performs its daily activities (Burt 1943; Rose 1982). HR size is a good indicator of the relationship between resource requirements and resource availability in the environment (Perry and Garland 2002; Hult and Germano 2015) and can be used to model several other processes, including an animal's cognitive map of its environment (Ousterhout and Burkhart 2017). HR size can be influenced by an individual's characteristics (e.g., size (Turner et al. 1969; Christian and Waldschmidt 1984; Van Sluys 1997), sex (Simonetti and Ortiz 1980; Rocha 1999; Robles and Halloy 2009), age (Davis and Ford 1983), social status (Jones and Droge 1980), sexual activity (Rose 1982; Stamps 1983), or trophic level (Christian and Waldschmidt 1984; Verwaijen and Damme 2008). Additionally, overlap can vary according to size, sex, and different classes (= hierarchies) of individuals (Stamps 1977; Kacoliris et al. 2009c). For example, in lizards, HR sizes are strongly associated with body size (Turner et al. 1969; Christian and Waldschmidt 1984) and depend, among other things, on an individual's dietary requirements and the cost of foraging (Escudero et al. 2020). The energetic hypothesis suggests that the size of the home range is related to body size and sex (Perry and Garland 2002), where larger species and males have wider home ranges (as compared to smaller species and females). In addition, lizards which are insectivores or carnivores have larger home ranges than herbivores (Christian and Waldschmidt 1984) as they need larger areas in which to search for and find their prey (Rocha 1999). Despite such a variety of factors influencing individual spatial use, it has been suggested that HR depends on habitat productivity and resource distribution, as they relate to individual energy requirements (Stellatelli et al. 2016a). Generally, lizards inhabiting highly complex habitats have smaller home ranges than those from homogeneous environments with relatively scarce resources (Perry and Garland 2002; Stellatelli et al. 2016a).

On the other hand, spatial ecology can also be understood from the microhabitat perspective. Differences in local abiotic conditions contribute to individual spatial use variation (Shaw 2020). In fact, animals are distributed in space in a non-random fashion, responding to microhabitat diversity of resources through selection strategies which increase their odds of survival (Ribeiro et al. 2009; Gómez Carella et al. 2019; He et al. 2019). The need to determine the selection or avoidance of a given microhabitat has been acknowledged as the first step in understanding ecological interactions between organisms and their surroundings (Kacoliris et al. 2009b). For example, one of the main factors determining microhabitat use in ectotherms is thermoregulation, as it affects their behavioral, physiological, and ecological characteristics (Huey 1982; Angilletta et al. 2009; Sinervo et al. 2010; Block et al. 2013). To maintain their body temperature within an adequate range in the face of environmental fluctuations, lizards move between microhabitats (Vicenzi et al. 2017; Gómez Alés et al. 2021). This selective exploitation of microhabitats allows them to maintain their body temperature and thus to optimize their physiological capacities, and, as a consequence, their ecological performance (Civantos 2000).

Reptiles are suitable organisms for studying the costs and benefits of spatial use. Since they are ectothermic vertebrates, their activity patterns depend mainly on environmental temperature and opportunities for thermoregulation (Sound and Veith 2011). In lizards, the physiological benefits of thermoregulation necessarily imply costs, such as energy expenditure, when moving between warm and cool spots or increased exposure to predators (Huey 1982; Block et al. 2013). Moreover, the time invested in thermoregulation cannot be used for other activities such as reproduction, feeding, social behaviors, and territory defense (Huey 1982; Vicenzi et al. 2019). Secondly, lizards show extensive intraspecific variation in patterns of space usage (Stamps 1983; Christian and Waldschmidt 1984) and, in consequence, have often been used as model systems in ecology (Huey et al. 1983; Vitt and Pianka 1994; Perry and Garland 2002). For example, sexual differences in thermoregulatory behavior have been observed, suggesting that males and females could differ in their thermal preferences (Maia-Carneiro and Rocha 2013a, 2013b) and, therefore, in the use of different microhabitats with dissimilar thermal characteristics (Block et al. 2013).

In South America, *Phymaturus* is a genus of liolaemid lizards that contains more than 60 species distributed throughout southwest (Lobo et al. 2016; Jara et al. 2019). However, the genus is conservative in many aspects of its life history and has a narrow ecological range (Corbalán and Debandi 2013; Vicenzi et al. 2017). All *Phymaturus* species are herbivorous, saxicolous, and viviparous, and inhabit cold and extreme environments of the Andes in Argentina and

Chile (Cruz et al. 2009; Lobo et al. 2016; Valdecantos et al. 2019). Moreover, the reproductive output of *Phymaturus* species is low; females have biennial reproductive cycles and small clutch sizes (n=2) (Boretto et al. 2007; Cabezas Cartes et al. 2010). In addition, most individuals within a species require between 7 and 9 years to reach sexual maturity (Boretto et al. 2015; Piantoni et al. 2006). These species, with limited dispersal capacities and confined to specialized habitats within restricted distribution ranges, are particularly vulnerable (Abdala et al. 2012) and susceptible to the effects of global warming (Ibargüengoytía and Casalins 2007; Sinervo et al. 2010; Vicenzi et al. 2017; Jara et al. 2019). Phymaturus williamsi is an endemic lizard of the Andes in Calingasta in the San Juan province of Argentina (Lobo et al. 2013) (Fig. 1a). This species has specialized morphological characteristics which allow it to live in rocky areas (Cei 1986). Despite vulnerability of the genus, P. williamsi is currently categorized as a species of "Least Concern" by the IUCN Red List of Threatened Species, with low priority for conservation (Abdala 2016). However, little is known about this species and the determinants of its spatial ecology remain an enigma.

For these reasons, here, we address the patterns involved in the variation of spatial ecology of *Phymaturus williamsi* in the Argentinean Puna. In particular, we determine the effect of intrinsic factors (sex, age, body size), microhabitat features (structure, sun exposure), microenvironment variables (substrate temperature, wind speed), and seasonality on home range size. We explore patterns of overlap among sexes and age classes. Additionally, we compare the microhabitat use of individuals among pre-reproductive, reproductive, and post-reproductive periods, addressing associations with microenvironmental variables and intrinsic factors. We hypothesize that the harsh conditions of the Puna habitat and this lizard's condition as herbivorous and saxicolous will limit spatial use by *P. williamsi*. Thus, these lizards will establish relatively smaller home ranges compared to other Liolaemidae. We also predict that *P. williamsi* males will have larger home ranges than females.

Materials and methods

Study area and fieldwork

We carried out this study in Quebrada Vallecito, located in the Andes Mountains, 40 km west of the town of Calingasta in the San Juan province of Argentina (31°11,021"S; 69°42,015"W, ~ 3000 m above sea level). The area has a cold and dry climate, with a mean annual temperature around 4.6 °C (annual temperature range 20.38 °C), a mean annual minimum temperature of -5.39 °C, and a mean annual maximum temperature of 14.99 °C; annual rainfall is 154 mm (Gómez Alés et al. 2021). In winter, accumulations of snow limit access to the site, so specimen collection can only take place from spring to early autumn (Cabezas Cartes et al. 2010). The area's vegetation is typical of that of the phytogeographic province of the Puna and is made up of shrublands of Artemisia mendozana, Fabiana denudata, Adesmia pinifolia, Adesmia horrida, Esphedra multiflora, and Esphedra breana, as well as cacti, such as Maihuenopsis glomerata or Lobivia formosa, and isolated grasslands of the genera Stipa and Javana in the understory (Márquez et al. 2016).

We conducted field work during March, October, and December 2011 and April 2012. This period includes spring, summer, and autumn, comprising the pre-reproductive, reproductive, and post-reproductive periods, respectively (Castro et al. 2018). The search for individuals took place for seven consecutive days between 10 am and 7 pm, the timeframe of lizard activity. Within the study area, we marked a grid (2500 m²) on an area with typical vegetation and rocky outcrops adequate for *P. williamsi* (Fig. 1b) and subdivided the grid into 5×5 m quadrants, measured with a laser rangefinder (Bushnell, Elite 1600, precision of ± 1 m)

Fig. 1 (a) Female adult *Phymaturus williamsi* in basalt rock crevice. (b) Representative image of one of the rocky outcrops used by the lizards in Quebrada Vallecito, San Juan, Argentina



(e.g., Robles and Halloy 2009). We assigned each vertex of the quadrant a number (columns) and a letter (row) (adapted from Gil et al. 1989). We performed sampling under all the weather conditions, except for rainy weather.

We performed random transects using the technique of visual encounter surveys (Heyer et al. 1994; Robles and Halloy 2009). Sampling was carried out by two observers standing 5 m apart (Frutos et al. 2007; Frutos and Belver 2007). Observations were made in independent unidirectional transects, avoiding visits to any location twice. In this way, we ensured that we are able to capture individual variability in spatial use while preventing repeated observations of the same individual (Pérez i de Lanuza and Carretero 2018). Using the mark-recapture method (Krebs 1999), we captured individuals by hand or noose on rocky outcrops and marked them with a toe-clipping numerical code for their permanent identification (Kacoliris et al. 2009a; Paulissen and Meyer 2000). We also marked the lizards with their corresponding identification number using non-toxic white paint (Grimm-Seyfarth et al. 2018) for their remote identification. For each record, we registered the following variables: date, spatial location on the grid, substrate temperature (with laser gun Benetech GM320), wind speed (measured with a Kestrel electronic anemometer), distance to nearest vegetation (where the maximum was a 60-cm radius from the capture site; using a tape measure), SVL (with digital caliper), and sex (determined based on external secondary characteristics) (Cei 1993). In addition, we classified individuals into adult or juvenile age classes, according to the shortest SVL recorded for reproductive individuals (females or males) available in the literature, as follows: female: 82.42 mm (82.42–101.78 mm; n = 18) and male: 83.48 mm (83.48–104.50 mm; n = 28); while juvenile SVL varied between 52.76 and 82.70 mm (n=21) (Cabezas Cartes et al. 2010). Subsequently, animals were released at the same capture site.

To record the spatial distribution of the lizards with respect to the characteristics of the microhabitat, we determined the type of substrate (feldspathic rock (Fr), basaltic rock (Br), or bare soil (Bs)). The rocks were differentiated by their mineralogy, type of fragmentation, and texture, which give them different thermal properties. We searched the three substrate types (Crump and Scott 1994) thoroughly, and when an individual was found, we recorded thermoregulating condition: whether sunlight shone on more (sun) or less (shade) than half of the body; shelter condition: whether the lizard had its entire body hidden in a shelter (crevice) or completely outside the shelter (exposed); and plant species found less than 60 cm from the capture site. In addition, substrate temperature and wind speed above ground were measured every time a lizard was spotted, using a laser gun (Benetech GM320) and an electronic anemometer (Kestrel). These microenvironmental variables are important to understand as they directly affect this lizard's thermoregulation. It was not possible to record data blindly because our study involved observation of focal animals in the field.

Statistical analyses

We estimated HR areas with the minimum convex polygon (MCP) (Rose 1982), suggested as the most appropriate method for herpetofauna (Winck et al. 2011) because it is easy to apply, it is less prone to home range overestimation, and it can be applied to small sample sizes (Olsen et al. 2011; Downs et al. 2012). The area within the polygon is an estimation of the space used by the individual (Rose 1982). We plotted and linked the capture points of the same individual, and this area represents the minimum HR. Because the MCP method is sensitive to outliers, their inclusion in the area can result in overestimates of true home ranges (Hult and Germano 2015). These outliers can result from exploratory displacements of an animal rather than a displacement needed for survival and reproduction or may even result from translocation by a predator bird, or from sampling errors. Following Hult and Germano (2015), we examined our data and found that only one of the sampled lizards had an atypical point (the home range of this lizard was 30 times larger than the average home range of the species). Therefore, we decided not to consider this exceptionally large home range and included 100% of the subsequent MCP for all HR estimations. We calculated and graphed the HR and overlap area between the HR of P. williamsi using Biotas software (Frutos et al. 2007; Kacoliris et al. 2009c) with a minimum of three observations per individual (Rose 1982). This minimum number of sightings was obtained following the Rose criteria (1982), which selects the number of points needed to describe 80% of the ultimate HR area. We work with the maximum number of sightings per individual, that is, one data point per individual in the analyses. Thus, we constructed data area curves using the number of sightings versus the accumulated mean home range for all individuals and determined that an average of 3 sightings explained 80% of the estimated home range size. Home range area increases with the number of sightings of an individual (Tinkle 1967; Rose 1982). This positive correlation disappears as the number of sightings increases and the home range of an individual becomes more completely described (Tinkle 1967; Rose 1982; Christian and Waldschmidt 1984). Thus, we estimate a minimum of 3 sightings as the smallest sample size at which the number of sightings was found to be noncorrelated with home range sizes (r=0.127; P=0.42) (Rose 1982; Stellatelli et al. 2016a).

We conducted all analyses and created all graphics with R statistical software, version 3.4.4 (R Core Team 2018). We tested assumptions of normality and homogeneity of variance with the one-sample Kolmogorov–Smirnov test and

with the Levene test, respectively. Means are given with ± 1 standard error (SE) and a significance of $\alpha = 0.05$ was used. We carried out parametric tests when data followed normality assumptions and homogeneity of variance, and nonparametric equivalent tests when such assumptions were not met (Sokal and Rohlf 1995; Crawley 2013). Additionally, we assessed multicollinearity using pairwise correlation tests, subsequently eliminating all the variables with correlation coefficients higher than 0.70 (Sokal and Rohlf 1995). We generated all the candidate models in the R package MASS (Bartoń 2013) and selected the best model (i.e., the most parsimonious) using the Akaike information criteria (AIC; see Burnham and Anderson 2004). All models were simplified using backward elimination of non-significant terms and model assumptions were tested in all cases. For each model, we evaluated data dispersion and the distribution of residual values (Crawley 2013).

To analyze the HR of these lizards, we performed two generalized linear models with a negative binomial distribution using the log link function. In the first model, HR size was the response variable, while sampling month, SVL, sex, and age were the predictor variables. In the second model, we tested five environmental predictor variables: wind speed, distance to vegetation, substrate type, thermoregulating condition, shelter condition, and substrate temperature. We eliminated the variable "shelter condition" since it was correlated (r > 0.70) with "thermoregulating condition" (Sokal and Rohlf 1995). To compare whether HR size for sex and age groups changed throughout the year (between months), we performed Kruskal–Wallis tests and subsequently carried out multiple comparison tests.

To analyze microhabitat use, only individuals without recaptures were considered. We then performed a third linear generalized model with a negative binomial distribution and log link function (Crawley 2013). We considered the number of individuals (counts) occupying each category of the variables. We eliminated the variable "thermoregulating condition" since it was correlated (r > 0.70) with "substrate temperature" (Sokal and Rohlf 1995). Lizard frequency was the response variable as a function of sampling month and sex, in relation to the following variables: wind speed, substrate temperature, substrate type, and shelter condition. To assess the presence of plant species in microhabitats, we performed chi-squared tests.

Results

Home range in relation to month, SVL, sex, and age

We obtained the HR size for 42 individuals throughout the entire study period (Fig. 2). Population average home range size was 8.12 m² (range: 0.0063–44.36, SE = 1.61 m²). For

these individuals, the mean number of sightings was 4.04 (range: 3–7, SE=1.06). Regarding sampling month, HR in March was significantly higher than in October (Z= -4.42, SE = 0.44, P < 0.0001), April (Z= -2.68, SE = 0.35, P=0.007), and December (Z= -2.40, SE=0.30, P=0.016). Following that trend, HR in April was significantly higher than in October (Z= -2.13, SE=0.48, P=0.033), but did not significantly differ from that of December (Z=0.63, SE=0.34, P=0.53) (Fig. 3a). HR size was not related to SVL (P>0.1). The best model for explaining such a variation had a weight of 50.75%.

The HR for females (mean = 3.98, SE = 0.82 m², n = 19) was significantly lower than that of juveniles (mean = 14.99, SE = 5.78 m², n = 8) (Z = 3.07, SE = 0.34, P = 0.002) and males (mean = 9.71, SE = 2.79 m², n = 15) (Z = 2.88, SE = 0.28, P = 0.004). No significant differences were found between males and juveniles (Z=0.692, SE=0.3554, P = 0.48872) (Fig. 3b). The HR for females varied significantly among sampling months ($H_{3,19} = 8.77$, P = 0.03). The test a posteriori showed that such differences occurred between March and October (Multiple K-W comparison test (M, O) = 2.89, P = 0.02) (Table 1). On the other hand, the mean HR size for males and juveniles did not significantly vary among sampling months ($H_{3,16} = 1.36$, P = 0.71; $H_{2,7} = 3.92$; P = 0.14, respectively).

Overlap

The mean HR overlap percentage was $4.73 \pm 2.07\%$. During the sampling months, the HR overlap percentage varied significantly (H=4.13, P=0.04), with higher percentages in March (mean=9.72, SE=4.37%, n=10) and lower percentages in October (mean=0.17, SE=0.15\%, n=2) (Fig. 2). Considering sexes and age class, we obtained 23 pairs of HR overlap. The highest percentage of HR overlap corresponded to the intersection between the HR of males and juveniles, followed by that between females. The HR overlap between the two sexes (M-F) was low in relation to the other values (1.05\%). The overlap between males was also low (0.80%) (Table 2). During breeding season (December to early April), males overlapped with up to three females, while during subsequent seasons, we only registered one intersexual overlap (Fig. 2).

Home range in relation to environmental variables

HR size in the different months was not dependent on wind speed, distance to vegetation, substrate type, thermoregulating condition, or shelter condition ($Z \ge 1.3$, P > 0.19 in all the cases). However, HR size exhibited a significantly positive relationship with substrate temperature (Z=3.58,



Fig. 2 Individual home ranges of adult females (circles), adult males (triangles), and juveniles (diamonds) of *Phymaturus williamsi* in Calingasta, San Juan, Argentina, between 2011 and 2012, as determined by the minimum convex polygon (MCP) method. Each lizard home

SE = 0.24, P = 0.0003) (Fig. 3c). The best model for explaining that variation had a weight of 23.65%.

Microhabitat use

The use of rocks was greater than 90% in all months (March: 92.5%, October: 100%, December: 95.9%, and

range is labeled with its identification number. The representation of the study area $(2500m^2)$ for each month is subdivided into a grid with 5×5 m quadrants. Each vertex of the quadrant is represented by a number (columns = *x*-axis) and a letter (rows = *y*-axis)

April: 95.2%). Except for during March, basalt rock was the microhabitat most used by lizards. In March and December, lizards were mainly found exposed to the sun, while in October, they were observed both in the sun and in the shade, and both exposed and in crevices in equal proportions. In April, they were more frequently found in the shade or in crevices (Table 3).

Fig. 3 Home range size of Phymaturus williamsi (between 2011 and 2012) in the Andes Mountain Range, Calingasta, Province of San Juan, Argentina. Representation of means and standard errors $(\pm SE)$: home range (a) according to month; (b) according to sex and age class. (c) Relationship between home range and substrate temperature, showing the trend and 95% confidence interval. The relationship was obtained using GLM with a negative binomial distribution



Table 1Snout-vent length and
home range size of *Phymaturus*
williamsi, sampled betweenMarch 2011 and April 2012 in
Calingasta, San Juan, Argentina

n Mean \pm SE Minimum Maximum March 34 96.73 \pm 2.84 58.00 116.00 Females 14 103.36 \pm 1.79 90.00 114.00 Males 11 107.63 \pm 1.92 97.00 116.00	Home ranges (m ²)			
March 34 96.73 ± 2.84 58.00 116.00 Females 14 103.36 ± 1.79 90.00 114.00 Males 11 107.63 ± 1.92 97.00 116.00	n	Mean ± SE	Minimum	Maximum
Females14 103.36 ± 1.79 90.00114.00Males11 107.63 ± 1.92 97.00 116.00	11	16.20 ± 4.99	1.68	44.36
Males 11 107.63+1.92 97.00 116.00	4	7.65 ± 0.73	6.32	9.73
	4	20.31 ± 9.10	1.68	44.36
Juveniles 9 68.78±2.21 58.00 77.00	3	28.5 ± 11.54	5.47	40.54
October 17 93.59 ± 4.30 56.00 115.00	7	1.89 ± 0.66	0.0063	4.73
Females 10 100.50±2.01 92.00 115.00	4	1.10 ± 0.75	0.0063	3.29
Males 3 107.66±4.10 100.00 114.00	1	4.73	-	-
Juveniles 4 65.75±5.73 56.00 78.00	2	2.07	1.34	2.80
December 42 87.597 ± 3.08 57.00 116.00	14	6.94 ± 1.42	0.50	18.05
Females 19 99.58±2.23 79.00 112.00	6	4.50 ± 1.72	0.05	12.47
Males 8 104.37±2.57 97.00 116.00	5	7.99 ± 2.16	5.14	16.59
Juveniles $15 63.47 \pm 1.63 57.00 77.00$	3	10.06 ± 4.39	2.88	18.05
April 27 96.63±2.91 53.00 115.00	10	5.26 ± 1.70	0.58	15.35
Females 16 100.81±2.19 81.00 110.00	5	2.71 ± 1.26	0.58	6.95
Males 7 104.00±2.39 96.00 115.00	5			15.25
Juveniles 4 67.00±5.55 53.00 80.00		7.82 ± 2.86	2.35	15.35

Mean \pm standard error (SE) and sample size (n) are presented

The best model for explaining the frequency of *P. williamsi* included 3 significant variables (substrate type, substrate temperature, and sex). The variables month, wind, and shelter conditions did not show significant interactions with the other variables. The use of basalt rock was higher than that of bare soil (Z = -4.10, SE = 0.33, P < 0.00001), but did not significantly differ from the use of feldspathic rock (Z = -0.58, SE = 0.16, P = 0.56). Following that trend, the use of feldspathic rock was higher than the use of bare

soil (Z=3.75, SE=0.34, P=0.0001). The frequency of males had a significant relationship with substrate temperature ($r^2 = 0.30$, P = 0.01), while the frequency of females and juveniles did not show a trend in relation to this variable ($r^2 = 0.0002$, P = 0.94; $r^2 = 0.002$, P = 0.86, respectively) (Fig. 4). Based on this, males were statistically different from females (Z=2.08, SE=0.20, P=0.037) and juveniles, albeit marginally non-significant (Z=-1.83, SE=0.24, P=0.066).

 Table 2
 Percentage overlap based on home ranges of *Phymaturus*

 williamsi in Calingasta (San Juan, Argentina) during 2011 and 2012

Overlap	п	Mean \pm SE	Minimum	Maximum
F-F	4	6.89 ± 6.49	0.01	26.34
M-M	3	0.80 ± 0.38	0.20	1.20
F-M	5	1.05 ± 0.44	0.0005	2.35
F-J	2	0.65 ± 0.33	0.32	0.98
M-J	6	9.02 ± 6.44	0.36	40.87
J-J	3	6.03 ± 4.72	0.07	15.35

Regarding the acronyms: Sample size (n), standard error (SE), and range of percentage overlap (Minimum–Maximum). Comparisons are for females (F), males (M), and juveniles (J)

Within the rock promontory, 41.1% of *P. williamsi* detections occurred in microhabitats with the presence of *E. multiflora*, followed by *Proustia cuneifolia* with a frequency of 21.4% (χ^2_{11} =539.76, *P*<0.0001). In all seasons, these two plant species were more frequent in microhabitats used by the lizards (Table 3) (March: χ^2_{11} =144.45, *P*<0.0001; October: χ^2_{11} =102.43, *P*<0.0001; December: χ^2_{11} =198.98, *P*<0.0001; and April: χ^2_{11} =140.39, *P*<0.0001).

Discussion

In this study, we define the spatial ecology of *P. williamsi* and its relationship to different variables for the first time. In accordance with our predictions, this species had the smallest HR as compared to other Liolaemidae: 20.7–1192.6 m² in *Phymaturus flagellifer* (Habit and Ortíz 1994; Celedón

Table 3Microhabitat use byindividuals (frequency in %)of *Phymaturus williamsi* inCalingasta, San Juan, Argentina(between 2011 and 2012)

Neghme et al. 2005), 22.3–59.8 m² in *L. lutzae* (Rocha 1999), 25.55–130.57 m² and 31.5–161.5 m² in *Liolaemus* quilmes (Halloy and Robles 2002; Robles and Halloy 2009), 40.7-140.4 m² in *Liolaemus koslowskyi* (Frutos and Belver 2007), 42.1–70.91 m² in *Liolaemus melanops* (Frutos et al. 2007), 54.96–39.87 m² in L. multimaculatus (Kacoliris et al. 2009c), 79.44-257.90 m² in Liolaemus espinozai (Cabrera and Scrocchi 2012), 37.8 m² in Liolaemus wiegmannii (Stellatelli et al. 2016b), and 63.8–387.3 m² in *Liolaemus* xanthoviridis (Escudero et al. 2020). This can be attributed to differences in diet, as carnivores (i.e., insectivores) have larger HR compared to omnivores and herbivores (Schoener 1968; Perry and Garland 2002; Namgail et al. 2014; Shaw 2020). P. williamsi is a generalist herbivore species (Castro et al. 2013), which would explain the smaller HR compared to other Liolaemus species with insectivore-omnivore diets (Rocha 1996, 1998; Perry and Garland 2002; Frutos et al. 2007; Kacoliris et al. 2009c; Stellatelli et al. 2016b). Additionally, productivity and resource distribution in the habitat are other factors which influence HR (Stellatelli et al. 2016a). Unlike the aforementioned Liolaemus, most of the *Phymaturus* species are restricted to isolated rocky promontories (Cei 1986, 1993). Consequently, the availability of rocky habitats with crevices limits the range of potential environments to be inhabited by this genus (Debandi et al. 2012; Vicenzi et al. 2017), which would explain the lower HR for P. williamsi. In turn, this could explain the differences with P. flagellifer, (herbivorous and saxicolous; Habit and Ortíz 1994), which inhabits a different Andes habitat, perhaps presenting differences in the size of the rocks, number of crevices, temperature, and food available. These

	March 2011	October 2011	December 2011	April 2012
Br	44.6	69.2	59.6	63.3
Fr	44.6	30.8	35.9	33.3
Bf	10.8	0	4.5	3.4
Sun	82.1	0.5	67.5	43.3
Shade	17.9	0.5	32.5	56.7
Exposed	89.3	0.5	70.8	33.3
Crevices	10.7	0.5	29.2	66.7
Adesmia pinifolia	5	0	0	1.5
Artemisia mendozana	0	5.1	2.6	12.5
Ephedra multiflora	43.7	43.9	36.8	35.9
Erodium cicutarium	1.2	0	0	0
Fabiana denudata	1.2	6.2	0	0
Graminea	18.7	8.2	18.4	10.9
Lobivia sp.	1.2	0	0	0
Mahuenopsis glomerata	1.2	0	0	0
Proustia cuneifolia	12.5	17.3	36.8	29.7
Schinus fasciculata	6.2	2.0	2.6	1.6
Sena aphila	1.2	6.1	0	0
Tetraglochin alatum	7.5	11.2	2.6	7.8

Comparisons are for basaltic rock (Br), feldspathic rock (Fr), bare floor (Bf), Sun/Shade condition, shelter condition (Exposed/Crevices), and plant species (found at less than 60 cm around the capture site)

Fig. 4 Relationship between frequency and substrate temperature (between sex and age classes) of *Phymaturus williamsi* between 2011 and 2012 in the Andes Mountain Range, Calingasta, Province of San Juan, Argentina. The shaded area represents 95% confidence interval. The relationships were obtained using GLM with a negative binomial distribution



factors are considered some of the most important for determining size of home range in *Phymaturus* (Habit and Ortiz 1994; Debandi et al. 2012).

Seasonal HR variation in P. williamsi could be explained in relation to seasonal climate changes (Rose 1982; Christian et al. 1983; Perry and Garland 2002; Sillero et al. 2016), since environmental temperatures strongly influence the capacity of lizards to patrol, monitor, or thermoregulate (Sound and Veith 2011). In ectotherms, when thermal availability is outside the preferred body temperature range, the cost of thermoregulation can be high and lizards must retreat to their shelters (Sinervo et al. 2010; Cabezas Cartes et al. 2014) to avoid risk of death. In fact, when temperatures are extreme, individuals decrease their HR due to their physiological restrictions (Ibargüengoytía et al. 2008). Thus, with the low temperatures (mean = 12.30, SE = 5.48 $^{\circ}$ C) and precipitation registered in October, lizards did not move and remained in their crevices most of the time. On the other hand, during March, higher HR probably results from energetic requirements (Schoener 1983; Christian and Waldschmidt 1984). Climatic conditions in high-mountain areas force this species to undergo long hibernation stages (Cabezas Cartes et al. 2010); in consequence, P. williamsi accumulates reserves to survive the winter, increasing its foraging area during late summer and early autumn (Wone and Beauchamp 2003).

On the other hand, it is widely accepted that individual size is related to HR size (e.g., Fox et al. 1981; Christian and Waldschmidt 1984; Perry and Garland 2002; Haenel et al. 2003); however, in *P. williamsi*, SVL was not related to HR size. This is in agreement with that reported for other Liolaemidae, such as *L. koslowskyi* (Frutos et al. unpublished results), *L. melanops* (Frutos et al. 2007), *L. multimaculatus* (Kacoliris et al. 2009c), *Liolaemus espinozai* (Cabrera and Scrocchi 2012), and *L. wiegmannii* (Stellatelli et al. 2016b) and raises questions about the generality of this hypothesis.

Social factors can also lead to great variations in lizards' use of space (Stamps 1977; Rose 1982; Schoener and Schoener 1982; Stellatelli et al. 2016a), e.g., due to individual differences in habitat familiarity (He et al. 2019). In this sense, as we expected, in *P. williamsi*, the average HR of males was more than twice that of females. However, males did not significantly differ from juveniles. Intersexual differences have also been observed in several lizard species (e.g., Turner et al. 1969; Rose 1982; Schoener and Schoener 1982; Stamps 1983; Rocha 1999; Perry and Garland 2002; Frutos and Belver 2007; Frutos et al. 2007; Germano 2007; Kacoliris et al. 2009c; Robles and Halloy 2009; Cabrera and Scrocchi 2012; Stellatelli et al. 2016a, b). In the genus *Phymaturus*, there is only information about intersexual HR size variations for P. *flagellifer* in Chile, with an average HR of 203.5–343.4 m² for males and 8.4-445.7 m² for females (Habit and Ortiz 1994). Intersexual HR differences might be explained by a combination of behavior, ecology, and physiology. For example, Miles et al. (2000) and Salido and Vicente (2019) propose that female lizards (especially those who are pregnant) run slower than males under a predation threat. Cooper et al. (1990) suggest that gravid lizards of some species achieve decreased conspicuousness by decreasing activity and remaining close to refuges. Thus, smaller HR in P. williamsi females might be explained by such scarce movement displacements, since remaining near crevices allows them to shelter easily. Similar results were reported for P. flagellifer females, which also had smaller HR (e.g., 8.4 m² HR in area 1), limited to only one rock (Habit and Ortiz 1994). On the other hand, it is likely that a combination of energy requirements and territoriality explains higher HR in males and juveniles as compared to female HR. Thus, dominant males might need HR which are large enough to meet their energy and mating demands (Simonetti and Ortiz 1980; Rocha 1999; Frutos and Belver 2007; Kacoliris et al. 2009c; Vidal et al. 2010; Halloy et al. 2013; Stellatelli et al. 2016a, b; He et al. 2019), while juveniles might have to travel greater distances in search of food as they are displaced from areas occupied by other dominant males.

Additionally, dominance hierarchies and territorial defense are related to the overlap between individuals (Kacoliris et al. 2009c). In populations with territorial behavior, low or no HR overlap is expected. Therefore, the low overlap observed in *P. williamsi* supports the idea of territorial defense in this species (Krekorian 1976; Rose 1982; Halloy and Robles 2002; Wone and Beauchamp 2003), mainly in males (overlap lower than 1%), with only two cases observed (one in March and one in April). This suggests exclusivity among males, who tend to defend exclusive areas in order to have access to females (following to Rocha 1999). This pattern has been reported in other lizard species, such as Tropidurus itambere (Van Sluys 1997), L. lutzae (Rocha 1999), Tropidurus torquatus (Ribeiro et al. 2009), L. quilmes (Robles and Halloy 2010), and Microlophus tigris (Ruiz et al. 2017). These results, in addition to a higher frequency of female-male overlap, a home range of larger size of males and a sex ratio biased towards females 2.03: 1 ($\chi^2_1 = 10.23$, P = 0.001) (AVE unpublished data), could indicate that P. williamsi has a polygynous mating system (following to Perry and Garland 2002). It would be interesting to address territoriality and its mating system in future studies.

Within their population, lizards can modify their HR seasonally. Male lizards, for example, can increase their HR during breeding season to find receptive females (e.g., Rose 1982; Stamps 1983; Rocha 1999, Shaw 2020). However, our results suggest that males maintained their HR, even during the warm season when greater mating activity would be expected (Castro et al. 2018). Females, instead, showed significant differences between March and October. This might have been caused by thermoregulatory behavior in response to thermal restrictions when temperatures were too low, as occurred in October, in agreement with that reported in Perry and Garland (2002). Our results showed the opposite trend of that found for L. quilmes, in which female HR did not vary throughout the two study years, while male HR was significantly higher during the reproductive period compared to the corresponding post-reproductive seasons (Halloy and Robles 2002). On the other hand, the higher HR overlap percentage in March agrees with that suggested by Cabezas Cartes et al. (2010) and Castro et al. (2018), who propose that mating in P. williams occurs in late summer and early autumn, for which a higher relative overlap would be expected in that period. In fact, during the summer, males overlapped with up to three females, while after mating season, we registered only one intersexual overlap.

On the other hand, HR overlap could also be related to parental care. However, in *P. williamsi*, no parental care has

been observed (Victorica and Acosta 2020). We recorded only two cases of overlap between females and juveniles (area less than 1 m2), dismissing the hypotheses of behavioral parental care, unlike other Liolaemidae (e.g., Phymaturus antofagastensis and P. punae, Cabezas Cartes et al. 2018; Phymaturus calcogaster and Liolaemus multicolor, Halloy et al. 2013; L. huacahuasicus, Halloy and Halloy 1997). However, some authors found a particular form of parental care through the presence of intra-abdominal yolk in newborn individuals of P. punae (Boretto et al. 2007), P. antofagastensis, Phymaturus cf. palluma (Cabezas Cartes et al. 2010), and Phymaturus zapalensis (Boretto and Ibargüengoytía 2009). This strategy might indicate a certain independence of juveniles at birth and could consequently explain the scarce overlap between females and juveniles detected in P. williamsi.

An interesting fact is that in *P. williamsi*, HR size was not related to the microenvironmental variables of wind speed, distance to vegetation, microhabitat used, thermoregulating condition, or shelter condition. However, HR size showed a positive relationship with substrate temperature. It is known that in ectotherms, the availability of adequate microhabitats for thermoregulation is one of the most important factors influencing activity patterns, habitat selection, and spatial distributions (Wilms et al. 2011). In this context, we believe that *P. williamsi* regulates its temperature seasonally, travelling between sites with different substrate temperatures according to its requirements, as do other *Phymaturus* species (Ibargüengoytía et al. 2008; Corbalán et al. 2013; Gómez Alés et al. 2017).

In Phymaturus, great uniformity in the use of space is observed, with an exclusive use of rocky habitats (e.g., Cei 1986, 1993; Espinoza et al. 2004; Ibargüengoytía 2005; Boretto and Ibargüengoytía 2006; Ibargüengoytía et al. 2008; Debandi et al. 2012; Vicenzi et al. 2017). However, recent studies showed that even in these habitats there are differences among species of the genus (Debandi et al. 2012). For example, certain species of the palluma group were positively related to sunlight availability, bare soil, isotherm, and slope (Corbalán and Debandi 2013). In the case of P. williamsi, it is specialized in two well-differentiated rock types (Br and Fr), using crevices as shelters to avoid lethal levels of sun radiation and predators (Fava and Acosta 2018; Fava et al. 2018). Bare soil as microhabitat was not utilized by lizards and might always be used as a passage site. These results are in agreement with those found by other authors who suggest that Phymaturus species prefer volcanic rocks (Debandi et al. 2012; Corbalán et al. 2013; Corbalán and Debandi 2014; Gómez Alés et al. 2017). The morphological characteristics of the genus, such as its flattened body and spiny tail, provide them with evolutionary advantages for sheltering in rocky crevices (Debandi et al. 2012). Thus, the availability of rocks with crevices might limit suitable habitat types for the species. We suggest that this hypothesis be taken into account in future studies on microhabitat use of *P. williamsi*.

Among the environmental microhabitat variables, we find that the number of individuals of P. williamsi was significantly related to substrate temperature. This is not surprising, as temperature is an important microhabitat variable for many lizards (e.g., Ibargüengoytía and Cussac 2002; Vidal et al. 2010; Gómez Alés et al. 2017; Stellatelli et al. 2017). It is known that P. williamsi, like other Phymaturus species (Vidal et al. 2010; Corbalán and Debandi 2013; Gómez Alés et al. 2017; Vicenzi et al. 2017, 2019), absorbs thermal energy from the substrate through thigmothermy and alternates this strategy with heliothermy according to the environmental thermic conditions (A. Laspiur unpublished data). However, we found intrapopulation variation in such relation, with only males showing a positive relationship with substrate temperature, differing from females not showing this trend. This might be due to a compromise in males between time assigned to social interactions and time assigned to thermoregulation, a situation that does not occur in females (Dunham et al. 1989). Males are exposed to higher temperatures, since they participate in territorial behaviors such as patrolling and controlling intruders (Vicenzi et al. 2019). In contrast, females avoid high temperatures and thermoregulate with higher precision to keep a stable temperature for the adequate development of embryos, for example (Gómez Alés et al. 2017; Stellatelli et al. 2018). This also occurs in P. patagonicus females, which have higher heat control than males, implying a physiological control on heat gain and loss (Ibargüengoytía 2005).

On the other hand, smaller lizards are more sensitive to heat exchange, since they have a low surface to volume ratio (Stellatelli et al. 2018). As such, P. williamsi juveniles might be more sensitive than males to overheating when substrate temperature is high, avoiding the risk of death through behavioral thermoregulation. This could explain the absence of a positive relationship between juveniles and substrate temperature. Likewise, juveniles of P. cf. palluma were more sensitive than adults to high summer temperatures (Eisenberg and Werning 2012). Another possibility is to consider thermal environment as a resource; thus, juveniles would be exposed to less thermally suitable microhabitats, due to their exclusion by dominant males from more thermally suitable microhabitats (Maia-Carneiro and Rocha 2013a; Delaney and Warner 2016). In fact, HR size of P. williamsi juveniles was similar to that of adult males, suggesting high displacement rates of juveniles which would expose them to thermal variations. Intraspecific differences in microhabitat use have been reported for other Liolaemidae, such as Phymaturus palluma, Liolaemus nigroviridis, and L. lutzae (Carothers et al. 1998; Vidal et al. 2010; Maia-Carneiro and Rocha 2013a, 2013b). However, for species of similar

environments, such as *Liolaemus parvus* and *Phymaturus extrilidus*, intraspecific differences in microhabitat use were not registered (Gómez Alés et al. 2017).

The P. williamsi lizards used microhabitats composed of combinations of 12 plant species of the 19 species recorded in the study area (Castro et al. 2013). However, the microhabitats used were characterized by the presence of only two dominant plant species: E. multiflora and P. cuneifolia. Like other species of lizards, P. williamsi seems to show a greater use of microhabitats with low vegetation (e.g., Vega 2001; Kacoliris et al. 2008, 2009b, 2010; Valdecantos et al. 2013; Peñalver Alcázar et al. 2016). Dias and Rocha (2004) suggest that the height of the vegetation in microhabitats has ecological implications related to body size and activity period which can facilitate the coexistence of sympatric species. Plants can provide protection against extreme surface temperatures (Kacoliris et al. 2008, 2010; Pietrek et al. 2009). In our study, it is possible that plants provide protection by reducing lizards' exposure to solar radiation. However, there is evidence that species specialized in certain habitats, such as Phymaturus, do not identify vegetation cover as a refuge due to their adaptation to the use of rocky crevices in the habitat (Pietrek et al. 2009). On the other hand, lizards can select plants according to their diet (Rocha et al. 2000; Beck and Jennings 2003; Corbalán and Debandi 2014), so P. williamsi could use microhabitats with the presence of these two plant species according to its food requirements. In fact, plants of the genus Ephedra are an important component of the diet of P. williamsi (Castro et al. 2013), as is the case of other species of *Phymaturus* (Debandi et al. 2012; Corbalán and Debandi 2014) such as P. palluma (Videla 1982) and P. antofagastensis (Acosta et al. 2008). It would be interesting to test these relationships in future studies.

We conclude that several factors interact to determine spatial ecology in this species, including individual characteristics (sex and age) and extrinsic factors such as seasonality, temperature (substrate), and habitat structure. Our data indicate that P. williamsi has the lowest home range among Liolaemidae recorded to date and uses highly restricted and specific microhabitats (basaltic and feldspathic rocks, with shrubby vegetation, avoiding open spaces with bare substrate or disaggregated soil). This rare lizard, with a small population size (AVE et al., unpublished data), occurs in delicate balance with its habitat, and as such any alteration of its environment exposes it to high vulnerability (Abdala et al. 2012). Therefore, we consider that this species could be highly vulnerable due to the combination of aspects analyzed in this study with its life history. In this sense, we consider that spatial ecology provides comprehensive information that must be taken into account for the conservation and management of species and their habitats, in light of the increasing fragmentation of habitats and increasing risks of extinction with progressively warming climates.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s00265-021-03120-2.

Acknowledgements We thank R. Nieva, M. Olmedo, S. Castro, and A. Laspiur for their assistance with the field work. Comments by M. Rodríguez, R. Nieva, and T. Martínez improved the manuscript. We thank two anonymous reviewers for valuable suggestions on the manuscript. Thanks also to Alyson Nuñez for assisting us with the English version. We also thank the Secretariat of Environment of the Province of San Juan for the permits issued (R056).

Author contribution All authors contributed to the study conception and design. AEVE and JCA collected data. AEVE and GAF performed the analyses. AEVE, GAF, and JCA wrote the manuscript.

Funding Financial support was received from the CICITCA scholarship (Consejo de Investigaciones Científicas y Técnicas y de Creación Artística, Res. 1117-R/11, AVE) and project CICITCA 881 (JCA), Argentina.

Data availability The datasets generated during and/or analyzed during the current study are available as electronic supplementary material.

Code availability Not applicable.

Declarations

Ethical approval This research was authorized by the Secretaría de Medio Ambiente, Dirección de Conservación y Áreas Protegidas, Provincia de San Juan (permits issued: R056) and complies with the ASIH/ HL/SSAR Guidelines for Use of Live Amphibians and Reptiles, as well as the regulations detailed in Argentinean National Law #14346. All protocols in the field, including the capture and handling of animals, were approved by the Department of Biology Committee of the National University of San Juan, Argentina.

Conflict of interest The authors declare no competing interests.

References

- Abdala CS, Acosta JL, Acosta JC et al (2012) Categorización del estado de conservación de las lagartijas y anfisbenas de la República Argentina. Cuad Herpetol 26:215–248
- Abdala S (2016) *Phymaturus williamsi*. The IUCN Red List of Threatened Species. https://doi.org/10.2305/IUCN.UK.2016-1.RLTS. T56252074A56252079.en
- Acosta JC, Villavicencio HJ, Blanco GM (2008) Phymaturus antofagastensis (NCN) Diet. Herpetol Rev 39:91
- Angilletta MJ, Sears MW Jr, Pringle RM (2009) Spatial dynamics of nesting behavior: Lizards shift microhabitats to construct nests with beneficial thermal properties. Ecology 90:2933–2939
- Bartoń K (2013) MuMIn: Multi-model inference R package version 1.9.5, http://CRAN.R-project.org/package=MuMIn. Accessed Apr 2019
- Beck DD, Jennings RD (2003) Habitat use by Gila monsters: the importance of shelters. Herpetol Monogr 17:111–129

- Block C, Stellatelli OA, García GO, Vega L, Isacch JP (2013) Factors affecting the thermal behavior of the sand lizard *Liolaemus wiegmannii* in natural and modified grassland of temperate coastal dunes from Argentina. J Therm Biol 38:560–569
- Boretto JM, Ibargüengoytía NR (2006) Asynchronous spermatogenesis and biennial female cycle of the viviparous lizard *Phymaturus antofagastensis* (Liolaemidae): reproductive responses to high altitudes and temperate climate of Catamarca, Argentina. Amphibia-Reptilia 27:25–36
- Boretto JM, Ibargüengoytía NR (2009) *Phymaturus* of Patagonia, Argentina: reproductive biology of *Phymaturus zapalensis* (Liolaemidae) and a comparison of sexual dimorphism within the genus. J Herpetol 43:96–104
- Boretto JM, Ibargüengoytía NR, Acosta JC, Blanco GM, Villavicencio HJ, Marinero JA (2007) Reproductive biology and sexual dimorphism of a high-altitude population of the viviparous lizard *Phymaturus punae* from the Andes in Argentina. Amphibia-Reptilia 28:1–7
- Boretto JM, Cabezas-Cartes F, Ibargüengoytía NR (2015) Energy allocation to growth and reproduction in a viviparous lizard endemic to the highlands of the Andes, Argentina. J Zool 297:77–86. https://doi.org/10.1111/jzo.12245
- Burnham KP, Anderson DR (2004) Multimodel inference: understanding AIC and BIC in model selection. Sociol Method Res 33:261–304
- Burt WH (1943) Territoriality and home range concepts as applied to mammals. J Mammal 24:346–352
- Cabezas Cartes F, Boretto J, Acosta JC, Jahn G, Blanco G, Laspiur A, Ibargüengoytía N (2010) Reproductive biology of *Phymaturus cf. palluma*: a vulnerable lizard from the highlands of the Andes, San Juan. Argentina Herpetol Conserv Biol 5:430–440
- Cabezas Cartes F, Kubisch EL, Ibargüengoytía NR (2014) Consequences of volcanic ash deposition on the locomotor performance of the *Phymaturus spectabilis* lizard from Patagonia, Argentina. J Exp Zool A 321:164–172
- Cabezas Cartes F, Boretto JM, Halloy M, Krenz JD, Ibargüengoytía NR (2018) Maternal behaviour in response to predation threats in a vulnerable lizard from Patagonia, Argentina. J Zool 304:175–181. https://doi.org/10.1111/jzo.12502
- Cabrera MP, Scrocchi GJ (2012) Áreas de acción en *Liolaemus espinozai* (Squamata: Liolaemidae) en Campo El Arenal, Catamarca, Argentina. Acta Zool Lilloana 56:54–65
- Carothers JH, Marquet PA, Jaksic FM (1998) Thermal ecology of a *Liolaemus* lizard assemblage along an Andean altitudinal gradient. Rev Chil Hist Nat 71:39–50
- Castro S, Laspiur A, Acosta JC (2013) Variación anual e intrapoblacional de la dieta de *Phymaturus cf. palluma* (Iguania: Liolaemidae) de los Andes centrales en Argentina. Rev Mex Biodiv 84:1258–1265. https://doi.org/10.7550/rmb.36050
- Castro S, Boretto JM, Blanco GM, Acosta JC (2018) Adjustment of the reproductive activity of vulnerable lizard *Phymaturus williamsi* at high altitudes. Herpetol Conserv Biol 13:283–293
- Cei JM (1986) Reptiles del Centro-Oeste y Sur de la Argentina. Monografie IV. Muzeo Regionale di Scienze Naturali Torino, Torino
- Cei JM (1993) Reptiles del noroeste, nordeste y este de la Argentina. Monografie XIV. Muzeo Regionale di Scienze Naturali Torino, Torino
- Celedón Neghme C, Salgado CR, Victoriano PF (2005) Preferencias alimentarias y potencial dispersor del lagarto herbívoro *Phymaturus flagellifer* (Tropiduridae) en los Andes. Gayana 69:266–276
- Christian KA, Waldschmidt S (1984) The relationship between lizard home range and body size: a reanalysis of the data. Herpetologica 40:68–75

- Christian KA, Tracy CR, Porter WP (1983) Seasonal shifts in body temperature and use of microhabitats by Galapagos land iguanas (Conolophus pallidus). Ecology 64:463–468
- Civantos E (2000) Home-range ecology, aggressive behaviour, and survival in juvenile lizards, *Psammodromus algirus*. Can J Zool 78:1681–1685
- Cooper WE Jr, Vitt LJ, Hedges R, Huey RB (1990) Locomotor impairment and defense in gravid lizards (Eumeces laticeps): behavioral shift in activity may offset costs of reproduction in an active forager. Behav Ecol Sociobiol 27:153–157
- Corbalán V, Debandi G (2013) Basking behaviour in two sympatric herbivorous lizards (Liolaemidae: *Phymaturus*) from the Payunia volcanic region of Argentina. J Nat Hist 47:1365–1378. https:// doi.org/10.1080/00222933.2012.759291
- Corbalán V, Debandi G (2014) Resource segregation in two herbivorous species of mountain lizards from Argentina. Herpetol J 24:201–208
- Corbalán V, Debandi G, Kubisch E (2013) Thermal ecology of two sympatric saxicolous lizards of the genus *Phymaturus* from the Payunia region (Argentina). J Therm Biol 38:384–389
- Crawley MJ (2013) The R book, 2nd edn. John Wiley and Sons, Chichester
- Crump ML, Scott NJ Jr (1994) Standard techniques for inventory and monitoring. Visual encounter surveys. In: Heyer WR, Donnelly MA, McDiarmid RW, Hayek AC, Foster MS (eds) Measuring and monitoring biological diversity. Standard methods for amphibians. Smithsonian Institution Press, Washington, pp 84–92
- Cruz F, Belver L, Acosta JC, Villavicencio HJ, Blanco G, Cánovas MG (2009) Thermal biology of *Phymaturus* lizards: evolutionary constraints or lack of environmental variation? Zoology 112:425–432
- Davis J, Ford RG (1983) Home range in the Western fence lizard, (Sceloporus occidentalis occidentalis). Copeia 1983:933–940
- Debandi G, Corbalán V, Scolaro JA, Roig Juñent SA (2012) Predicting the environmental niche of the genus *Phymaturus*: are *palluma* and *patagonicus* groups ecologically differentiated? Austral Ecol 37:392–400. https://doi.org/10.1111/j.1442-9993.2011.02295.x
- Delaney DM, Warner DA (2016) Age- and sex-specific variations in microhabitat and macrohabitat use in a territorial lizard. Behav Ecol Sociobiol 70:981–991. https://doi.org/10.1007/ s00265-016-2121-3
- Dias EJR, Rocha CFD (2004) Thermal ecology, activity patterns, and microhabitat use by two sympatric whiptail lizards (*Cnemidophorus abaetensis* and *Cnemidophorus ocellifer*) from Northeastern Brazil. J Herpetol 38:586–588
- Downs JA, Heller JH, Loraamm R, Stein D, Mcdaniel C, Onorato D (2012) Accuracy of home range estimators for homogeneous and inhomogeneous point patterns. Ecol Model 225:66–73
- Dunham AE, Grant BW, Overall KL (1989) Interfaces between biophysical and physiological ecology and the population ecology of terrestrial vertebrate ectotherms. Physiol Zool 62:335–355
- Eisenberg T, Werning H (2012) *Phymaturus cf. palluma* in captivity: observations on its reproduction and biology. Salamandra 48:198–206
- Escudero PC, González Marín MA, Morando M, Avila LJ (2020) Use of space and its relationship with sex, body size, and color polymorphism in *Liolaemus xanthoviridis* (Iguania: Liolaemini) in Patagonia. J Herpetol 54:57–66
- Espinoza RE, Wiens JJ, Tracy CR (2004) Recurrent evolution of herbivory in small, cold-climate lizards: Breaking the ecophysiological rules of reptilian herbivory. P Natl Acad Sci USA 48:16819–16824
- Evans LC, Sibly RM, Thorbek P, Sims I, Oliver TH, Walters RJ (2019) Integrating the influence of weather into mechanistic models of butterfly movement. Mov Ecol 7:24. https://doi.org/10.1186/ s40462-019-0171-7

- Fava G, Acosta JC (2018) Escape distance and escape latency following simulated rapid bird attacks in an Andean lizard, Phymaturus williamsi. Behaviour 155:1–21. https://doi.org/10.1163/15685 39X-00003506
- Fava G, Acosta JC, Victorica AE, Martínez T, Rodríguez M (2018) *Phymaturus williamsi*. Predator-Prey Interaction. Herpetol Rev 49:331–332
- Fox SF, Rose E, Myers R (1981) Dominance and the acquisition of superior home ranges in the lizard Uta stansburiana. Ecology 62:888–893
- Frutos N, Belver LC (2007) Dominio vital de *Liolaemus koslowskyi* Etheridge (1993) (Iguania: Liolaemini) en el noroeste de la provincia de La Rioja, Argentina. Cuad Herpetol 21:83–92
- Frutos N, Camporro LA, Avila LJ (2007) Ámbito de hogar de *Liolaemus melanops* Burmeister, 1888 (Squamata: Liolaemini) en el Centro de Chubut, Argentina. Gayana 71:142–149
- Germano JM (2007) Movements, home ranges, and capture effect of the endangered Otago skink (*Oligosoma otagense*). J Herpetol 41:179–186
- Gil M, Pérez Mallado V, Guerrero F (1989) Estimación de dominios vitales en anfibios y reptiles. Metodología de muestreo y análisis de datos. Rev Esp Herp 3:275–286
- Gómez Alés R, Acosta JC, Laspiur A (2017) Thermal biology in two syntopic lizards, *Phymaturus extrilidus* and *Liolaemus parvus*, in the Puna region of Argentina. J Therm Biol 68:73–82
- Gómez Alés R, Acosta JC, Valdez F, Martínez TA, Acosta R, Rodriguez Muñoz MJ, Corrales FR, L, (2021) Comparative thermal ecophysiology in *Pristidactylus scapulatus* populations from the Puna region of Argentina. Zoology 145:125903. https://doi.org/ 10.1016/j.zool.2021.125903
- Gómez Carella D, Speziale KL, Lambertucci SA (2019) Estado del conocimiento en ecología y conservación de los roquedales de la Argentina: Una revisión. Ecol Austral 29:315–328
- Grimm-Seyfarth A, Mihoub J, Gruber B, Henle K (2018) Some like it hot: from individual to population responses of an arboreal aridzone gecko to local and distant climate. Ecol Monogr 88:336– 352. https://doi.org/10.1002/ecm.1301
- Habit EM, Ortiz JC (1994) Ámbito de hogar de *Phymaturus flagellifer* (Reptilia, Tropiduridae). Bol Soc Biol Concepc 65:149–152
- Haenel GJ, Smith LC, John Alder HB (2003) Home range analysis in Sceloporus undulates: a test of spatial relationships and reproductive success. Copeia 2003:113–123
- Halloy M, Halloy S (1997) An indirect form of parental care in a highaltitude viviparous lizard, *Liolaemus huacahuasicus (Tropiduridae)*. Bull Md Herpetol Soc 33:139–155
- Halloy M, Robles C (2002) Spatial distribution in a neotropical lizard *Liolaemus quilmes* (Liolaemidae): site fidelity and overlapping among males and females. Bull Md Herpetol Soc 38:118–129
- Halloy M, Robles C, Salica MJ, Semhan R, Juárez Heredia V, Vicente N (2013) Estudios de comportamiento y ecología de lagartijas de los géneros *Liolaemus* y *Phymaturus* (Iguania: Liolaemini). Cuad Herpetol 27:15–26
- He P, Maldonado-Chaparro AA, Farine DR (2019) The role of habitat configuration in shaping social structure: a gap in studies of animal social complexity. Behav Ecol Sociobiol 73:9. https://doi. org/10.1007/s00265-018-2602-7
- Heyer R, Donnelly M, Diarmid R, Hayek L, Foster M (1994) Measuring and monitoring biological diversity. Smithsonian Institution Press, Washington, Standard methods for amphibians
- Hoffmann A, Abt Tietje G, Reyer H-U (2015) Spatial behavior in relation to mating systems: movement patterns, nearest-neighbor distances, and mating success in diploid and polyploid frog hybrids (*Pelophylax esculentus*). Behav Ecol Sociobiol 69:501–517. https://doi.org/10.1007/s00265-014-1862-0
- Huey RB (1982) Temperature, physiology, and the ecology of reptiles. In: Gans C, Pough FH (eds) Biology of the Reptilia, vol 12.

Physiological Ecology. Academic Press. London, Physiology C, pp 25–91

- Huey RB (1991) Physiological consequences of habitat selection. Am Nat 137:91–115
- Huey RB, Pianka ER, Schoener TW (1983) Lizard ecology: studies of a model organism. Harvard University Press, Cambridge
- Huey RB, Peterson CR, Arnold JS, Porter WP (1989) Hot rocks and not-so-hot rocks: retreat-site selection by garter snakes and its thermal consequences. Ecology 70:931–944
- Hult S, Germano DJ (2015) Habitat use and home range of *Phrynosoma blainvillii* in the San Joaquin Desert of California. Herpetol Conserv Biol 10:850–863
- Ibargüengoytía NR (2005) Field, selected body temperature and thermal tolerance of the syntopic lizards *Phymaturus patagonicus* and *Liolaemus elongates* (Iguania: Liolaemidae). J Arid Environ 62:435–448
- Ibargüengoytía NR, Casalins L (2007) Reproductive biology of the southernmost gecko *Homonota darwini*: convergent life-history patterns among southern hemisphere reptiles living in harsh environments. J Herpetol 41:71–79
- Ibargüengoytía NR, Cussac V (2002) Body temperatures of two viviparous *Liolaemus* lizard species, in Patagonian rain forest and steppe. Herpetol J 12:131–134
- Ibargüengoytía NR, Acosta JC, Boretto JM, Villavicencio HJ, Marinero JA, Krenz JD (2008) Field thermal biology in *Phymaturus* lizards: comparisons from the Andes to the Patagonian steppe in Argentina. J Arid Environ 72:1620–1630
- Jara M, García Roa R, Escobar LE, Torres Carvajal O, Pincheira Donoso D (2019) Alternative reproductive adaptations predict asymmetric responses to climate change in lizards. Sci Rep 9:5093. https://doi.org/10.1038/s41598-019-41670-8
- Jones SM, Droge DL (1980) Home range size and spatial distributions of two sympatric lizard species (*Sceloporus undulatus, Holbrookia maculata*) in the Sandhills of Nebraska. Herpetologica 36:127–132
- Kacoliris FP, Williams J, Sánchez Véliz G, Rafael A (2008) Observaciones sobre el uso de cavidades en la arena por parte de la lagartija de los médanos (*Liolaemus multimaculatus*). Cuadernos De Herpetología 22:87–89
- Kacoliris FP, Berkunsky I, Williams J (2009a) Methods for assessing population size in Sand Dune Lizards (*Liolaemus multimaculatus*). Herpetologica 65:219–226
- Kacoliris FP, Celsi CE, Monserrat AL (2009b) Microhabitat use by the sand dune lizard *Liolaemus multimaculatus* in a pampean coastal area in Argentina. Herpetol J 19:61–67
- Kacoliris FP, Williams JD, Ruiz de Arcaute C, Cassino C (2009c) Home range size and overlap in *Liolaemus multimaculatus* (Squamata: Liolaemidae) in Pampean coastal dunes of Argentina. S Am J Herpetol 4:229–234
- Kacoliris FP, Williams JD, Molinari A (2010) Selection of key features of vegetation and escape behavior in the Sand Dune Lizard (*Liol-aemus multimaculatus*). Anim Biol Leiden Neth 60:157–167
- Krebs CJ (1999) Ecological methodology, 2nd edn. Benjamin Cummings, Menlo Park
- Krekorian CO (1976) Home range size and overlap and their relationship to food abundance in the desert iguana, *Dipsosaurus dorsalis*. Herpetologica 4:405–412
- Lobo F, Laspiur A, Acosta JC (2013) Description of new Andean species of the genus *Phymaturus* (Iguania: Liolaemidae) from Northwestern Argentina. Zootaxa 3683:117–132
- Lobo F, Barrasso DA, Hibbard T, Basso NG (2016) On the evolution and diversification of an Andean clade of reptiles: combining morphology and DNA sequences of the palluma group (Liolaemidae: *Phymaturus*). Zool J Linn Soc 176:648–673
- Maia-Carneiro T, Rocha C (2013a) Influences of sex, ontogeny and body size on the thermal ecology of *Liolaemus lutzae*

(Squamata, Liolaemidae) in a restinga remnant in southeastern Brazil. J Therm Biol 38:41–46

- Maia-Carneiro T, Rocha C (2013b) Seasonal variations in behaviour of thermoregulation in juveniles and adults *Liolaemus lutzae* (Squamata, Liolaemidae) in a remnant of Brazilian restinga. Behav Process 100:48–53
- Márquez J, Martinez Carretero E, Dalmasso A (2016) Provincias fitogeográficas de la provincia de San Juan. In: Martinez Carretero E, García A (eds) San Juan Ambiental. CONICET, Mendoza, pp 187–197
- Miles DB, Sinervo B, Frankino WA (2000) Reproductive burden, locomotor performance, and the cost of reproduction in free ranging lizards. Evolution 54:1386–1395
- Namgail T, Takekawa J, Balachandran S, Sathiyaselvam P, Mundkur T, Newman SH (2014) Space use of wintering waterbirds in India: Influence of trophic ecology on home-range size. Curr Zool 60:616–621
- Olsen J, Downs JA, Tucker T, Trost S (2011) Home-range size and territorial calling of Southern Boobooks (*Ninox novaeseelandiae*) in adjacent territories. J Raptor Res 45:136–142
- Ousterhout BH, Burkhart JJ (2017) Moving beyond the plane: measuring 3D home ranges of juvenile salamanders with passive integrated transponder (PIT) tags. Behav Ecol Sociobiol 71:59. https://doi.org/10.1007/s00265-017-2284-6
- Paulissen MA, Meyer HA (2000) The effect of toe-clipping on the gecko *Hemidactylus turcicus*. J Herpetol 34:282–285
- Peñalver Alcázar M, Aragon P, Breedveld MC, Fitze PS (2016) Microhabitat selection in the common lizard: implications of biotic interactions, age, sex, local processes, and model transferability among populations. Ecol Evol 6:3594–3607. https:// doi.org/10.1002/ece3.2138
- Pérez i de Lanuza G, Carretero MA (2018) Partial divergence in microhabitat use suggests environmental-dependent selection on a colour polymorphic lizard. Behav Ecol Sociobiol 72:138. https://doi.org/10.1007/s00265-018-2550-2
- Perry G, Garland T (2002) Lizard home ranges revisited: effects of sex, body size, diet, habitat, and phylogeny. Ecology 83:1870–1885
- Pianka ER (1986) Ecology and natural history of desert lizards: analyses of the ecological niche and community structure. Princeton University Press, Princeton, NJ
- Piantoni C, Ibargüengoytía NR, Cussac VE (2006) Age and growth of the Patagonian lizard *Phymaturus patagonicus*. Amphibia-Reptilia 27:385–392
- Pietrek AG, Walker RS, Novaro AJ (2009) Susceptibility of lizards to predation under two levels of vegetative cover. J Arid Environ 73:574–577
- Pough FH, Andrews RM, Cadle JE, Crump ML, Savitzky AH, Wells KD (2001) Herpetology, 2nd edn. Prentice Hall, New York
- R Core Team (2018) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. http://www.R-project.org/
- Ribeiro LB, Sousa BM, Gomides SC (2009) Range structure, microhabitat use, and activity patterns of the saxicolous lizard *Tropidurus torquatus* (Tropiduridae) on a rock outcrop in Minas Gerais, Brazil. Rev Chil Hist Nat 82:577–588
- Robles C, Halloy M (2008) Seven-year relative abundance in two syntopic neotropical lizards, *Liolaemus quilmes* and *L. ramirezae* (Liolaemidae), from northwestern Argentina. Cuad Herpetol 22:73–79
- Robles C, Halloy M (2009) Home ranges and reproductive strategies in a neotropical lizard, *Liolaemus quilmes* (Iguania: Liolaemidae). S Am J Herpetol 4:253–258
- Robles C, Halloy M (2010) Core areas overlap in a neotropical lizard *Liolaemus quilmes*: relationship with territoriality and reproductive strategy. Herpetol J 20:243–248

- Rocha CFD (1996) Seasonal shift in lizard diet: the seasonality in food resources affecting the diet of *Liolaemus lutzae* (Tropiduridae). Ciênc Cult 48:264–269
- Rocha CFD (1998) Ontogenetic shift in the rate of plant consumption in a tropical Lizard (*Liolaemus lutzae*). J Herpetol 32:274–279
- Rocha CFD (1999) Home range of the Tropidurid lizard *Liolaemus lut*zae: sexual and body size differences. Rev Bras Biol 59:125–130
- Rocha CFD, Vrcibradic D, Araújo AF (2000) Ecofisiologia de répteis de restingas brasileiras. In: Esteves FA, Lacerda FA (eds) Ecologia de restingas e lagoas costeiras, Rio de Janeiro. Universidade Federal do Rio de Janeiro, Rio de Janeiro, pp 117–149
- Rose B (1982) Lizard home ranges: methodology and functions. Herpetologica 16:253–269
- Ruiz J, Junes K, Pérez J (2017) Ámbito de hogar de la lagartija de las lomas Microlophus tigris (Sauria: Tropiduridae) en las Lomas de Carabayllo Lima-Perú. Rev Peru Biol 24:357–362. https://doi. org/10.15381/rpb.v24i4.14070
- Salido CA, Vicente NS (2019) Sex and refuge distance influence escape decision in a *Liolaemus* lizard when it is approached by a terrestrial predator. Behaviour 156:909–925. https://doi.org/10.1163/1568539X-00003546
- Schoener TW (1968) Sizes of feeding territories among birds. Ecology 49:123–141
- Schoener TW (1983) Simple models of optimal feeding-territory size: a reconciliation. Am Nat 121:608–629
- Schoener TW, Schoener A (1982) Intraspecific variation in home-range size in some *Anolis* lizards. Ecology 63:809–823
- Shaw AK (2020) Causes and consequences of individual variation in animal movement. Mov Ecol 8:12. https://doi.org/10.1186/ s40462-020-0197-x
- Sillero N, Corti C, Carretero MA (2016) Home ranges of parthenogenetic and bisexual species in a community of *Darevskia* lizards (Reptilia: Lacertidae). Zool Middle East 62:306–318. https://doi. org/10.1080/09397140.2016.1257403
- Simonetti J, Ortiz JC (1980) Dominio en *Liolaemus kuhlmani* (Reptilia: Iguanidae). Anu Mus Hist Nat Valparaíso 13:167–172
- Sinervo B, Méndez de la Cruz F, Miles DB et al (2010) Erosion of lizard diversity by climate change and altered thermal niches. Science 328:894–899
- Sokal RR, Rohlf FJ (1995) Biometry: the principles and practice of statistics in biological research, 3rd edn. WH Freeman and Company, New York
- Sound P, Veith M (2011) Weather effects on intrahabitat movements of the western green lizard, *Lacerta bilineata* (Daudin, 1802), at its northern distribution range border: a radiotracking study. Can J Zool 78:1831–1839
- Stamps JA (1977) Social behavior and spacing patterns in lizards. In: Gans C, Tinkle DW (eds) Biology of the Reptilia, vol 7A. Academic Press, New York, pp 256–334
- Stamps JA (1983) Sexual selection, sexual dimorphism, and territoriality. In: Huey RB, Pianka ER, Schoener TW (eds) Lizard ecology: studies of a model organism. Harvard University Press, Cambridge, pp 169–204
- Stellatelli OA, Block C, Moreno Azocar DL, Vega LE, Isacch JP, Cruz FB (2016a) Scale dependency of *Liolaemus* lizards' home range in response to different environmental variables. Curr Zool 62:521–530. https://doi.org/10.1093/cz/zow021
- Stellatelli OA, Block C, Vega LE, Isacch JP, Cruz FB (2016b) Factors affecting the spatial ecology of the lizard *Liolaemus wiegmannii* in the pampasic coastal dunes of Argentina. Herpetol J 26:11–19
- Stellatelli OA, Block C, Villalba A, Vega LE, Dajil JE, Cruz FB (2017) Behavioral compensation buffers body temperatures of

two *Liolaemus* lizards under contrasting environments from the temperate Pampas: a Bogert effect? Ethol Ecol Evol 30:1–22. https://doi.org/10.1080/03949370.2017.1388293

- Stellatelli OA, Villalba A, Block C, Vega LE, Dajil JE, Cruz FB (2018) Seasonal shifts in the thermal biology of the lizard *Liolaemus tandiliensis* (Squamata, Liolaemidae). J Therm Biol 73:61–70
- Tinkle DW (1967) The life and demography of the side-blotched lizard, *Ilta stanisburiana*. Misc Pub Mus Zool Univ Michigan 132:1–182
- Turner FB, Jennrich RI, Weintraub JD (1969) Home ranges and body sizes of lizards. Ecology 50:1076–1081
- Valdecantos S, Martínez V, Lobo F, Cruz FB (2013) Thermal biology of *Liolaemus* lizards from the high Andes: being efficient despite adversity. J Therm Biol 38:126–134
- Valdecantos S, Lobo F, Perotti MG, Moreno Azócar DL, Cruz FB (2019) Sexual size dimorphism, allometry and fecundity in a lineage of South American viviparous lizards (Liolaemidae: *Phymaturus*). Zool Anz 279:152–163. https://doi.org/10.1016/j. jcz.2019.02.003
- Van Sluys M (1997) Home range of the saxicolous lizard *Tropidurus itambere* in Southeastern Brazil. Copeia 1997:623–628
- Vega L (2001) Herpetofauna: diversidad, ecología e historia natural. In: Iribarne O (ed) Reserva de Biosfera Mar Chiquita: Características Físicas, Biológicas y Ecológicas. Editorial Martín, Mar del Plata, pp 213–226
- Verwaijen D, Damme RV (2008) Wide home ranges for widely foraging lizards. Zoology 111:37–47
- Vicenzi N, Corbalán V, Miles D, Sinervo B, Ibargüengoytía N (2017) Range increment or range detriment? Predicting potential changes in distribution caused by climate change for the endemic high-Andean lizard *Phymaturus palluma*. Biol Conserv 206:151–160
- Vicenzi N, Ibargüengoytía N, Corbalán V (2019) Activity patterns and thermoregulatory behavior of the viviparous lizard *Phymaturus palluma* in Aconcagua Provincial Park, Argentine Andes. Herpetol Conserv Biol 14:337–348
- Victorica AE, Acosta JC (2020) *Phymaturus williamsi* Parturition Behaviour Herpetol Rev 51:131
- Vidal M, Habit E, Victoriano P, González Gajardo A, Ortiz JC (2010) Thermoregulation and activity pattern of the high-mountain lizard *Phymaturus palluma* (Tropiduridae) in Chile. Zoológica 27:13–18
- Videla F (1982) Observaciones etológicas preliminares sobre *Phymaturus palluma* (Reptilia, Iguanidae) del roquedal precordillerano mendocino. Bol Mus Cs Nat Antrop Moyano 3:57–62
- Vitt LJ, Pianka ER (1994) Lizard ecology: historical and experimental perspectives. Princeton University Press, Princeton, NJ
- Wilms TM, Wagner P, Shobrak M, Rödder D, Böhme W (2011) Living on the edge? On the thermobiology and activity pattern of the large herbivorous desert lizard Uromastyx aegyptia microlepis Blanford, 1875 at Mahazat as-Sayd Protected Area, Saudi Arabia. J Arid Environ 75:636–647
- Winck GR, Blanco CC, Cechin SZ (2011) Population ecology of *Tupi-nambis merianae* (Squamata, Teiidae): home-range, activity and space use. Anim Biol 61:493–510
- Wone B, Beauchamp B (2003) Movement, home range and activity patterns of the horned lizard *Phrynosoma mcallii*. Herpetologica 37:679–686

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.